



Article The Ground-Based Absolute Radiometric Calibration of the Landsat 9 Operational Land Imager

Jeffrey S. Czapla-Myers ^{1,*}, Kurtis J. Thome ², Nikolaus J. Anderson ¹, Larry M. Leigh ³, Cibele Teixeira Pinto ³ and Brian N. Wenny ⁴

- Remote Sensing Group, Wyant College of Optical Sciences, University of Arizona, Tucson, AZ 85721, USA; nanderson@optics.arizona.edu
- NASA Goddard Space Flight Center, Code 618, Greenbelt, MD 20771, USA; kurtis.thome@nasa.gov
 Office of Engineering Research College of Engineering South Datas State University
- Office of Engineering Research, College of Engineering, South Dakota State University,
- Brookings, SD 57007, USA; larry.leigh@sdstate.edu (L.M.L.); cibele.teixeirapinto@sdstate.edu (C.T.P.)
- ⁴ Science Systems & Applications Inc., Lanham, MD 20706, USA; brian.n.wenny@nasa.gov
- * Correspondence: jscm@optics.arizona.edu; Tel.: +1-520-621-4242

Abstract: This paper presents the initial vicarious radiometric calibration results for Landsat 9 OLI using a combination of ground-based techniques and test sites located in Nevada, California, and South Dakota, USA. The field data collection methods include the traditional reflectance-based approach and the automated Radiometric Calibration Test Site (RadCaTS). The results for top-of-atmosphere spectral radiance show an average ratio (OLI/ground measurements) of 1.03, 1.01, 1.00, 1.02, 1.02, 1.01, 0.98, and 1.01 for Landsat 9 OLI bands 1–8, which is within the design specification of $\pm 5\%$ for spectral radiance. The results for top-of-atmosphere reflectance show an average ratio (OLI/ground measurements) of 0.99, 0.99, 1.00, 1.02, 1.01, 1.02, 1.00, and 1.00 for Landsat 9 OLI bands 1–8, which is within the design specification of $\pm 3\%$ for top-of-atmosphere reflectance.

Keywords: Landsat 9; OLI; radiometric calibration; absolute calibration; radiometric validation



Citation: Czapla-Myers, J.S.; Thome, K.J.; Anderson, N.J.; Leigh, L.M.; Pinto, C.T.; Wenny, B.N. The Ground-Based Absolute Radiometric Calibration of the Landsat 9 Operational Land Imager. *Remote Sens.* 2024, *16*, 1101. https://doi.org/ 10.3390/rs16061101

Academic Editor: Pablo Rodríguez-Gonzálvez

Received: 31 January 2024 Revised: 4 March 2024 Accepted: 14 March 2024 Published: 21 March 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

1. Introduction

The Landsat program is the longest-running Earth observation satellite program in history, providing a continuous record of the Earth's land surface since the launch of the first Landsat satellite, Landsat 1, in 1972. The program is a joint effort between NASA and the United States Geological Survey (USGS) and currently consists of three operational satellites as follows: Landsat 7, Landsat 8, and Landsat 9. The Landsat program has had a significant impact on various fields, including agriculture, forestry, geology, and land use planning. It has been used to monitor natural disasters, track urban growth, and map global land cover changes. In addition, the program has facilitated the development of numerous applications and technologies, including Geographic Information Systems (GISs), remote sensing, and image processing techniques.

Over the years, the Landsat satellites have undergone several improvements in terms of technology, data quality, and coverage. The Landsat 1–3 satellites were equipped with a Multispectral Scanner (MSS) that had four spectral bands. Landsat 4 and 5 carried the Thematic Mapper (TM) instrument, which provided more detailed and accurate data with seven spectral bands. Landsat 7 launched in 1999 with the Enhanced Thematic Mapper Plus (ETM+) instrument, which added a panchromatic band, improved geolocation accuracy, and onboard calibration. Today, the Landsat program continues to provide valuable data for scientific research, resource management, and decision making at local, national, and global scales. Its legacy is a testament to the importance of long-term Earth observation programs for understanding and managing the planet.

The most recent additions to the program are Landsat 8 and Landsat 9, launched in 2013 and 2021, respectively. They carry essentially identical payloads, which include

advanced multispectral sensors that operate in the visible and near-infrared (VNIR), shortwave infrared (SWIR), and thermal infrared (TIR) spectral regions. Additional upgrades include improved spatial resolution and data quality. Landsat 9 is the most recent satellite in the continuous fifty year history of the Landsat program, which will continue with the Landsat Next program [1,2]. It was launched on 27 September 2021 from the Vandenberg Space Force Base using a United Launch Alliance Atlas V 401 launch vehicle and placed into a sun-synchronous 705 km orbit with a sixteen day repeat cycle. It is effectively in the same equatorial crossing time as Landsat 7 before the lowering of its orbit, which is part of its decommissioning phase.

The Landsat 9 payload consists of the same two instruments on Landsat 8 as follows: the Operational Land Imager (OLI) and the Thermal Infrared Sensor (TIRS), which provide complimentary spectral coverage. The OLI operates in the solar-reflective spectral region (400 nm to 2500 nm) using nine spectral channels, while the TIRS is a two-band thermal sensor that operates from 10.6 μ m to 12.5 μ m. The Landsat 9 OLI spectral bands retain the same spatial characteristics as those of Landsat 8 OLI and Landsat 7 ETM+, which is a 30 m ground sampling distance (GSD) for the multispectral bands, a 15 m GSD for the panchromatic band, and a 185 km swath width (Table 1). The most notable change in recent Landsat solar-reflective bands is the transition from a whiskbroom sensor to a pushbroom sensor (ETM+ vs. OLI), which allows for a higher signal-to-noise ratio (SNR). ETM+ has an 8-bit radiometric resolution, while the Landsat 9 OLI is that of a 14-bit radiometric resolution, which allows for a higher detection change over dark targets such as vegetation and water [3,4]. The relative spectral responses (RSRs) for the OLI multispectral bands 1–7 are shown in Figure 1.

Landsat 9 OLI					
Band	Center Wavelength (nm)	Bandwidth (FWHM, nm)	GSD (m)		
1	443	16	30		
2	483	60	30		
3	561	57	30		
4	655	37	30		
5	865	29	30		
6	1609	86	30		
7	2201	189	30		
8 (pan)	592	172	15		

Table 1. The spectral bands and GSD of the Landsat 9 OLI. The center wavelength defined here is the band average of the relative spectral response (RSR) for each band.

The requirement for the absolute radiometric uncertainty of the Landsat 9 OLI is the same as that for the Landsat 8 OLI as follows: $\pm 5\%$ for top-of-atmosphere (TOA) spectral radiance and $\pm 3\%$ for TOA reflectance (both at the k = 1 level). The Landsat 9 OLI onboard radiometric calibration system is identical to that of Landsat 8 OLI, and the various subsystems include a shutter for dark measurements, two solar diffusers (working and pristine), two stimulation lamp assemblies, and the ability to view the moon through a spacecraft maneuver [2,5]. As with the Landsat 8 OLI, the Landsat 9 OLI underwent a rigorous pre-launch calibration and characterization process [6–10].

The radiometric calibration and validation of Earth-observing sensors continues to be an essential component of post-launch operation. Monitoring the spectral, radiometric, and geometric temporal changes in a sensor is critical for the scientific community, who rely on these data for a wide variety of scientific, environmental, and societal applications. Ground-based techniques are now in a mature state and continue to be used to monitor and characterize any changes that may occur during the operational life of a sensor. The value of these systems is their independence from any onboard calibration systems that may also change over time. Independent radiometric validation has also become more important with the growing number of operational satellite constellations, many of which consist of CubeSats or nanosats that typically lack onboard calibration systems. Current ground-based techniques include automated instrumented sites [11], in situ measurements by on-site personnel [12–18], pseudo-invariant calibration sites (PICSs) [19], lunar observations, and mirror-based artificial targets [20,21]. Cross-calibration with suitable spaceborne and airborne sensors is also widely used for calibration and validation studies. The independent validation of Earth-observing systems ensures that data from sensors with different spectral, spatial, and temporal characteristics can be placed on the same traceable radiometric scale, thereby allowing for harmonized data products with a well-defined uncertainty.



Figure 1. The relative spectral response (RSR) for the Landsat 9 OLI multispectral bands 1–5 (**left**) as well as 6 and 7 (**right**). The panchromatic band 8 is omitted for clarity, and the cirrus band 9 is also excluded since it is not used in this work.

Vicarious radiometric calibration is a ground-based technique that has been in use for almost 40 years [22–24]. It provides an independent analysis that measures and monitors any post-launch changes in the airborne or spaceborne system under test. This work presents the results from the initial nineteen months after the launch of the Landsat 9 OLI and includes measurements using on-site personnel as well as those using an instrumented automated test site. The Landsat 9 OLI results from the tandem flight phase of Landsat 8 and 9 are also presented. Section 2 of this paper presents the reflectance-based and RadCaTS ground-based methodologies. Section 3 describes the field campaigns during the nineteen-month period of this work (November 2021 to June 2023), and Section 4 summarizes the results for each of the ground collects. Section 5 presents a brief description of the ground measurement uncertainty, and Section 6 provides a summary of the results and conclusions of this work.

2. Methodology

2.1. Ground-Based Vicarious Radiometric Calibration

In general terms, ground-based vicarious radiometric calibration uses scenes of known spectral radiance and compares them to a sensor under a test to evaluate its absolute radiometric calibration. In the solar-reflective regime, the creation of a scene with known spectral radiance requires knowledge of the surface reflectance [25] and the atmospheric transmittance [18], as well as knowledge of the atmospheric scattering properties. The surface reflectance is typically measured by ground instrumentation that may or may not require on-site personnel. Atmospheric measurements are made using suitable spectroradiometers, with two examples being the Automated Solar Radiometer (ASR) [26–28] and the Cimel CE-318 solar–lunar photometer [29,30]. In general terms, the results of these two measurements are used in a radiative transfer code to determine the at-sensor spectral radiance for a given airborne or spaceborne instrument [31]. The work described in this paper uses two independent ground-based techniques, the first of which uses a system of auto-

mated instruments to make surface reflectance and atmospheric measurements (RadCaTS), while the other uses ground-based personnel for similar measurements (reflectance-based approach). Each of these systems is described in the next two sections, with the RadCaTS being introduced first since it is now the primary data collection technique used by the University of Arizona (UArizona).

2.2. Automated Measurements: The Radiometric Calibration Test Site (RadCaTS)2.2.1. RadCaTS Development

The RadCaTS was developed by the Remote Sensing Group of the Wyant College of Optical Sciences at the University of Arizona (UArizona) in the early 2000s [32–34]. During the prototyping phase, the RadCaTS was used to supplement the in situ data that were routinely collected by on-site personnel using the traditional reflectance-based approach. A data processing methodology was developed, tested, and compared to the reflectance-based results during the prototyping stage. The knowledge gained in this process resulted in the development of radiometrically stable, all-weather, multispectral ground-viewing radiometers (GVRs), the first examples of which were deployed in 2012 [35,36]. Additional upgrades over the past ten years have included a satellite uplink station, upgraded Cimel CE-318T solar–lunar photometer [29], and a GVR with linear motion [37]. The RadCaTS has been used successfully for such sensors as the Landsat 7 ETM+ [38], the Landsat 8 and 9 OLIs [17,39,40], Terra and Aqua MODIS [41], ASTER [42,43], SNPP, NOAA-20 VIIRS [44], Sentinel-2A and -2B MSI [38], and GOES-16 and -17 ABI [45,46]. It has also been used to validate the calibration of airborne sensors [15,47,48].

The expansion from on-site personnel to an automated system evolved as a sensible response to the growing number of national and international Earth-observing satellite programs. In addition to national space programs, the creation of nanosat and CubeSat constellations by commercial companies is also increasing rapidly. Automated ground-based radiometric calibration sources become increasingly important to the operators of these programs since many of these small satellites have limited or no onboard radiometric calibration systems [49–58].

2.2.2. RadCaTS: Part of a Global Radiometric Calibration Network

The RadCaTS is one of the four original sites in the Radiometric Calibration Network Working Group (RadCalNet), which was established by the Committee on Earth Observing Satellites (CEOS) Working Group on Calibration and Validation (WGCV) Infrared and Visible Optical Sensors (IVOS) [11]. There are currently five instrumented sites located in the USA, France, Namibia, and China, with new sites currently under construction and evaluation [59]. TOA reflectance, bottom-of-atmosphere (BOA) reflectance (or the surface BRF, as denoted in this work), and various atmospheric parameters from all sites are all freely available to registered users at the RadCalNet web portal (www.radcalnet.org, accessed on 1 January 2024). RadCalNet data are currently being used to validate the calibration of a variety of spaceborne and airborne sensors [15,47,60–70]. The web portal also contains documentation for those interested in developing a RadCalNet site, information for the users of RadCalNet data, and historical intercomparison reports from a RadCalNet user workshop. Additionally, the portal hosts the RadCalNet Forum, which allows users to interact directly with the RadCalNet Working Group, including site owners and data providers (forum.radcalnet.org, accessed on 1 January 2024). Notices regarding site status, data availability, and data quality are provided by site operators on a regular basis in order to keep users informed of data outages or quality issues.

2.2.3. Development of Custom Ground-Viewing Radiometers for RadCaTS

In 2002, the RadCaTS concept started with a one-channel proof-of-concept radiometer that used a light-emitting diode (LED) as a detector. The decision to use an LED as a detector was influenced by the successful Global Learning and Observations to Benefit the Environment (GLOBE) program, which promotes the collection and analysis of environmental data

as well as the collaboration of students around the world [71–74]. The preliminary RadCaTS radiometer was based on the GLOBE sun photometer, and prototypes were constructed, characterized, and field-tested using various VNIR LEDs. Modest improvements to the radiometer design over the next few years included the addition of two more channels in the VNIR region, as well as powered optics to better control the field of view [75]. The proof-of-concept instruments allowed the data processing architecture to be developed and tested, and it was determined that to be a truly operational and well-calibrated radiometric resource for the Earth observation community, new radiometers had to be developed.

In 2012, the prototype three-channel VNIR GVRs were replaced with a new version developed by UArizona to transition to a truly operational and autonomous test site. Improvements include a temperature-controlled focal plane, interference filters for spectral selection, and a solar power supply for remote operation. The eight spectral channels of the GVRs used at the RadCaTS are nominally centered at 400 nm, 450 nm, 500 nm, 550 nm, 650 nm, 850 nm, 1000 nm, and 1550 nm, and they have 20 nm bandwidths (full width at half maximum, FWHM). The focal plane of each GVR is mounted 1.5 m above the ground, which corresponds to a spot-size diameter on the ground of 27 cm, as defined by the 10° full field of view for each channel. The GVRs were designed, developed, tested, and characterized at UArizona specifically for the spectral and spatial characteristics of the surface at Railroad Valley.

In the original operational phase in 2012, there were three GVRs deployed to the Rad-CaTS, and in 2014, a fourth one was deployed. The development of more GVRs continued throughout the years, along with the addition of a satellite communication network that allows data to be uploaded daily to UArizona for processing. There are currently seven GVRs at the RadCaTS in a nadir-viewing configuration. In the past, two GVRs were mounted in the viewing configuration of the GOES-East and GOES-West geostationary satellites for work with the Advanced Baseline Imager (ABI), but they were reverted to nadir-viewing in 2020 [45,46]. The positions and number of the GVRs required to obtain the same uncertainty in the surface bidirectional reflectance factor (BRF) as the reflectance-based approach were assessed using high-resolution satellite imagery [33,75–77]. Examples of the current GVRs at the RadCaTS are shown in Figure 2, which also shows a recent prototype design that uses a new mount for the radiometer head. The new mount allows for 90 cm of linear motion, which increases the spatial sampling. The current positions of the seven GVRs are shown in Figure 3.



Figure 2. (a) An example of two GVRs at the RadCaTS. Currently, six of the seven have a static mount. (b) A GVR with a linear translation mount that provides 90 cm of motion.



Figure 3. The current locations of the seven RadCaTS GVRs (21–25, 41, and 42), the meteorological station (Met), and Cimel sun photometer(s) (Cimel) at Railroad Valley. GVR 23 has 90 cm of linear motion. The feature running north–south is a gravel road.

2.2.4. Railroad Valley, Nevada, USA

The RadCaTS is located at Railroad Valley, Nevada, USA, which has a usable area of 15 km \times 15 km at an altitude of 1435 m. Preliminary work by UArizona in the 1990s sought to identify suitable ground test sites in the Southwestern United States for the reflectance-based approach and also for the cross calibration of Earth-observing sensors that operate in the solar-reflective regime (400 nm to 2500 nm) [78]. Examples of the principal criteria for a suitable site included the following:

- Surface reflectance: >0.3, with the aim to reduce uncertainties in the path radiance.
- Spatial uniformity: to reduce uncertainties due to sensor misregistration during cross calibration studies.
- Large size: to reduce uncertainties from adjacency effects.
- Arid region: to reduce surface reflectance changes due to precipitation and/or the presence of clouds.
- High altitude: to reduce uncertainties in atmospheric characterization due to aerosols.
- Accessibility: in the 1990s, UArizona deployed a mobile lab pulled by a truck during field campaigns, so the test site had to be accessible with the truck and trailer.

A preliminary exploratory trip to Railroad Valley occurred in 1996, and regular field work commenced soon after. Other sites used by UArizona include Lunar Lake, Nevada; [79,80], Ivanpah Playa, California; and White Sands Missile Range, New Mexico [81]. By early 2000, Railroad Valley and Ivanpah Playa were the two main sites used by the group for sensors such as Landsat 5 TM, Landsat 7 ETM+, Aqua and Terra MODIS, and ASTER. There were three main collection areas used by UArizona at Railroad Valley as follows: one for pushbroom sensors (e.g., ASTER), one for whiskbroom sensors (e.g., TM and ETM+), and one for large-footprint sensors such as MODIS. The original concept for the RadCaTS design specified that it was to be suitable for moderate-resolution sensors such as MODIS, so the large-footprint area at Railroad Valley was chosen as the region of interest (ROI) for

the RadCaTS. The choice to use a preexisting area was also important because it meant that a library of surface BRF measurements was already being established.

The RadCaTS ROI is 1 km \times 1 km in size and centered at 38.497° latitude, -115.690° longitude. In terms of the World Reference System-2 (WRS-2), which is a global notation system for Landsat data, it is path 40 and row 33. Figure 4 shows a Landsat 9 OLI image of Railroad Valley collected on 10 July 2022 at 18:21 UTC. The 1 km² RadCaTS ROI shown in Figure 3 is indicated by the yellow box in Figure 4.



Figure 4. A Landsat 9 OLI image of Railroad Valley (10 July 2022, 18:21 UTC) created using OLI bands 4, 3, and 2. The yellow box denotes the 1 km \times 1 km RadCaTS ROI (WRS-2 path 40, row 33).

2.2.5. Atmospheric Measurements at the RadCaTS

The atmospheric measurements at the RadCaTS are performed using a Cimel sun photometer that follows the AERONET measurment collection protocol [29,30,82–84]. Throughout the day, direct solar irradiance and sky radiance measurements are made automatically, and the data are uploaded hourly through a satellite uplink to the NASA Goddard Space Flight Center (GSFC) for further processing, where they are subsequently made available for public access on the AERONET web portal. In the case of the RadCaTS, the atmospheric quantities required as the input to the radiative transfer code include the following:

- Aerosol optical depth (AOD);
- Angstrom exponent;
- Columnar water vapor;
- Columnar ozone;
- Carbon dioxide.

The AOD, Angstrom exponent, and columnar water vapor data are obtained from AERONET, while the columnar ozone data are obtained from either the Ozone Monitoring Instrument (OMI) or the Ozone Mapping and Profiler Suite (OMPS) data portals. The daily carbon dioxide amount is downloaded from the NOAA Global Monitoring Laboratory (GML) web portal (gml.noaa.gov, accessed on 1 February 2024).

Once these data have been acquired, they are used as input into the radiative transfer code to determine the spectral transmission of the direct solar beam, as well as the diffuse sky irradiance. The MODTRAN mid-latitude summer atmospheric profile and corresponding standard atmosphere vertical profile column amounts are used for the Rad-CaTS. The exoatmospheric solar irradiance model used for Landsat work is Chance–Kurucz (ChKur) [85]. The aerosol optical properties are determined using an internal database of Mie-generated aerosol phase functions in MODTRAN. The aerosol optical properties at the RadCaTS are assumed to follow a power-law size distribution that has a given Angstrom exponent and complex refractive index. The use of such an assumption at desert sites such as Railroad Valley has been justified because of the typical low amount of aerosol loading coupled with a relatively high surface BRF. The MODTRAN output of interest for the RadCaTS is the atmospheric spectral transmittance from 350 nm to 2500 nm as well as the diffuse sky irradiance, E_{sky} , which is used in the determination of the surface BRF, described in the following section.

2.2.6. Surface Reflectance Measurements at the RadCaTS

One of the primary differences between the RadCaTS and the traditional reflectancebased approach is that of the methodology used to determine the surface BRF. The RadCaTS uses GVRs to measure reflected radiance from the test site surface, while the reflectancebased approach uses the ratio of digital numbers (DNs) collected with a portable spectroradiometer over the ground to those collected over a reference panel [18,42,86–91]. The most notable difference is that the RadCaTS requires a solar exoatmospheric irradiance model to convert the measured reflected surface radiance to a surface BRF [17].

Operationally, the BRF of the RadCaTS ROI is determined for a given time of interest by initially determining the BRF in all eight channels of each GVR. This step requires the spectral radiance measurement in each GVR channel, the atmospheric measurements made by the Cimel sun photometers, the ancillary data (e.g., ambient temperature and pressure), and an exoatmospheric irradiance model. These data are processed in the radiative transfer model (currently MODTRAN 6 [31]), and the output is the multispectral surface BRF, an example of which is shown in Figure 5 for a Landsat 9 overpass on 24 June 2022.





The final step in processing the surface reflectance measurements is to convert the multispectral surface BRF to a hyperspectral BRF for use once again in the radiative transfer code. The conversion is completed using a least-squares fit of the multispectral GVR data to one of the hyperspectral data sets in a reference library, which is a collection of portable spectroradiometer measurements made by UArizona personnel over the past thirty years. UArizona personnel are the main source for these data, but more recent BRF data have been obtained from NASA JPL personnel who perform similar measurements for different sensors [48,92,93]. A limited amount of snow BRF data have been collected by UArizona personnel during various field campaigns at Railroad Valley, but they are generally not used in RadCaTS or RadCalNet processing. An example of the resulting conversion from

multispectral to hyperspectral BRF is shown in Figure 6. The procedure to determine the surface BRF for a given date and time is summarized below and also in Figure 7. The equation for the surface BRF determination is shown in Appendix A.



Figure 6. The average multispectral BRF retrieved from the seven GVRs in Figure 5, including the fit to one of the hyperspectral portable spectroradiometer data in the reference library.



Figure 7. RadCaTS surface BRF processing.

- 1. Determine the surface BRF in each GVR channel.
 - Calculate the spectral radiance measured by each GVR channel.
 - Use a reference monthly average BRF for the diffuse sky irradiance (E_{sky}) calculation.

- Obtain the processed AERONET data for the time of interest, including the AOD_{500nm}, precipitable water vapor (WV) and Angstrom exponent.
- Download atmospheric data such as ozone and CO₂ amount.
- Download ancillary data such as ambient temperature and barometric pressure from the on-site meteorological station at Railroad Valley.
- Use the AERONET measurements, the atmospheric data (CO₂ and O₃), and ambient temperature and pressure data as input into a radiative transfer code.
- 2. Convert the multispectral GVR surface BRF to a hyperspectral BRF.
 - Compute the average surface BRF for each of the eight GVR bands in order to obtain one multispectral surface BRF for the RadCaTS ROI.
 - Perform a least-squares best fit of the multispectral surface BRF to a library of reference BRF values obtained with multispectral spectroradiometers.
- 3. Compare the output hyperspectral surface BRF with the one used in 1b.
 - If the average difference is higher than a predetermined value, rerun step 1 using the new hyperspectral surface BRF.
 - Continue this process until the difference between the two values converge to being within the predetermined value. (Note: the wavelength regions used for this comparison are as follows: 400 nm to 1200 nm, 1500 nm to 1700 nm, and 2000 nm to 2250 nm. These spectral regions are chosen in order to avoid absorption regions in the atmosphere.)
- 4. At this point, the hyperspectral surface BRF has been determined for the given time and date of interest.

2.2.7. Determination of TOA Spectral Radiance and TOA Reflectance

Once the hyperspectral surface BRF has been determined at the RadCaTS for the date and time of interest, the TOA spectral radiance and reflectance are determined using one final run of the radiative transfer code. The resulting output is the hyperspectral TOA spectral radiance (and reflectance, if applicable), which is then band-averaged to the bands of the sensor under testing. In the final step, these values are compared to the sensor imagery, which is downloaded from the USGS Earth Explorer web portal in the case of Landsat. An example of the hyperspectral TOA spectral radiance determined at the RadCaTS for the Landsat 9 overpass on 24 June 2022 is shown in Figure 8, and a summary of the processing scheme is shown in Figure 9.



Figure 8. TOA spectral radiance at RadCaTS on 24 June 2022 at 18:20 UTC for the Landsat 9 OLI viewing conditions.



Figure 9. RadCaTS TOA spectral radiance (TOA reflectance) processing.

2.3. On-Site Personnel: The Reflectance-Based Approach

2.3.1. Overview

The reflectance-based approach to vicarious radiometric calibration is a well-understood technique that has been used successfully over the past 35 years for airborne and spaceborne sensors [12–14,18,81,88,94–97]. Like the RadCaTS process, measurements of the atmosphere and surface BRF are made during an overpass time of interest, and the results are used in a radiative transfer code to determine the TOA spectral radiance (or reflectance). The results are then compared to the imagery for the sensor under test.

2.3.2. Field Test Sites

The test site used by UArizona and the NASA GSFC during the brief tandem flight phase of Landsat 8 and Landsat 9 is Ivanpah Playa, California, USA, which continues to be used by a variety of national and international research groups for post-launch radiometric calibration work. As with Railroad Valley, temporal in situ data collection over the past 35 years have allowed UArizona to create a temporal library of hyperspectral surface reflectance data for Ivanpah Playa. It is located southwest of Las Vegas, Nevada, on the Nevada–California border, and it has a useable area of approximately 2 km × 10 km at an altitude of 800 m (Figure 10a). Traditionally, the area east of the Interstate-15 highway has been used for data collection, but this work uses the area to the west of the highway, which was chosen primarily due to easier accessibility.

South Dakota State University (SDSU) has historically used vegetated sites located near Brookings, South Dakota, USA (Figure 10b). The three sites used for this work are 150 m \times 250 m, 120 m \times 180 m, and 120 m \times 180 m in size, at an average altitude of 500 m. The grass at each site is routinely maintained throughout the field campaign season, which typically ranges from April to October.

The reflectance-based approach is the predecessor to the RadCaTS, and the in situ measurements required to obtain TOA quantities follow the same in principle as follows: measurements of the surface BRF and the atmospheric transmission during overpass are required as inputs to a radiative transfer code to determine the TOA quantities of interest, which are then compared to the airborne or spaceborne sensor under test. Thorough descriptions of the atmospheric and surface BRF measurements have been previously published [12,86,94,95,98], but a general description relevant to the Landsat 9 work presented here is included.



Figure 10. (a) Landsat 9 OLI image of Ivanpah Playa on 13 November 2021, 18:19 UTC (WRS-2 path 40, row 35); (b) Landsat 9 OLI image of Brookings, South Dakota on 27 June 2022, 17:11 UTC (WRS-2 path 29, row 29). Each true color image was created using OLI bands 4, 3, and 2. The yellow arrows in each image denote the approximate location of the ground sites.

2.3.3. Atmospheric Measurements

The atmospheric characterization required during a vicarious calibration field campaign has traditionally been performed using a suite of custom-built and commercial instruments that temporarily operate onsite throughout the time of interest. The goal of such measurements is to determine the atmospheric transmittance, which is used in a radiative transfer code to determine the TOA spectral radiance for a specific time, date, as well as illumination and view geometry.

The research groups in this work make base measurements of solar irradiance extinction due to absorption and scattering with a ten-channel multispectral automated solar radiometer (ASR) that uses narrow bandwidth interference filters for spectral selection. The ten spectral channels are centered at 380 nm, 400 nm, 441 nm, 520 nm, 611 nm, 670 nm, 780 nm, 870 nm, 940 nm, and 1030 nm [26]. The ASR tracks the sun and measures the incoming solar irradiance extinction caused by atmospheric absorption and scattering. The data are then used to derive the total atmospheric spectral optical depths, which can be further broken down into subcomponents such as molecular, aerosol, ozone, and water vapor optical depths [99,100]. In desert sites used by UArizona, the aerosols are modeled by a power-law distribution, and the Angstrom exponent is used to define the aerosol size distribution. Columnar water vapor is determined through a modified Langley approach [27,28], while the columnar ozone amount is obtained from the OMI or OMPS. Ancillary data such as temperature and pressure are measured using additional equipment brought to the site. Following a manual setup procedure that includes alignment with the sun, the ASR tracks the sun throughout the day through a quadrant cell that is coregistered to the optical axis of the radiometer. Measurements are typically made every minute during the surface BRF data collection process, which is described in the next section. The spectral optical depths are determined using a retrieval based on the Langley method [101]. The optical depth results are used in an inversion method to determine the ozone optical depth and an aerosol size distribution parameter [100]. As mentioned, the work presented here assumes that the aerosol size distribution can be defined by a power law. The main advantage of this assumption is that it only requires one value, known as the Angstrom turbidity parameter, to define the aerosol size distribution.

In the final step of ASR data processing, the optical depths are determined from 350 nm to 2500 nm for use as inputs to the radiative transfer code. The columnar water vapor amount is derived from the ASR data using a modified Langley approach [102]. The Angstrom exponent determined using ASR measurements is also used to compute the Mie scattering phase functions, which are also used in the radiative transfer code.

SDSU uses ASR measurements in conjunction with Langley analyses to derive instantaneous optical depth values. The initial model is propagated from the sun through the atmosphere to ground level, predicting spectral radiance at the reflectance standard, spectral radiance at the grass target, and the diffuse-to-global irradiance ratio of the sky. The predicted results are compared to the radiance measurements made with the portable spectroradiometer, while the diffuse-to-global ratio is compared to a Yankee Environmental Systems shadow-band radiometer to identify any anomalies in the model predictions. The model is then updated and rerun until a fit to reality is obtained.

2.3.4. Surface Reflectance Measurements

The surface BRF is typically measured using a portable hyperspectral spectroradiometer having a wavelength range of 350 nm to 2500 nm, and it is carried across the site in a suitable predetermined pattern. The foreoptic is mounted to a boom arm to ensure shadowfree measurements. Periodic measurements of a Spectralon reference panel are made at predetermined points throughout the site collection, and the output digital numbers (DNs) over the two targets (Spectralon and ground) are ratioed to determine the hyperspectral surface BRF for each measurement. Routine measurements of the UArizona and SDSU Spectralon panel BRFs as a function of wavelength and angle are made at the UArizona goniometric laboratory to monitor changes caused by regular field use, including ultraviolet (UV) light and dust exposure [89–91,103]. In the final BRF processing step, the individual transect measurements are averaged together to create one hyperspectral surface BRF to be used in the radiative transfer code for the given sensor overpass.

In addition to the typical hyperspectral spectroradiometer used to collect surface reflectance data, a recently developed multispectral instrument was deployed by GSFC personnel during the November 2021 field campaign. It is known as the Calibration Test Site SI-Traceable Transfer Radiometer (CaTSSITTR), and it was developed by UArizona after the RadCaTS became operational and with the increasing involvement of UArizona in RadCalNet. The main goal in the development of the CaTSSITTR is to allow one-person operation in the field while maintaining an SNR and absolute accuracy on par with laboratory transfer radiometers. There are currently two identical CaTSSITTR instruments as follows: CaTSSITTR-A (Arizona) and CaTSSITTR-G (Goddard) [104–107], and their design specifications mirror those of a RadCaTS GVR, minus the SWIR channel at 1550 nm. The CaTSSITTR measurements at Ivanpah Playa were modeled on the reflectance-based approach, but instead of walking the site taking continuous measurements between panel measurements, the instrument was moved to eight randomly predetermined locations within the overall test site. Operation of a portable spectroradiometer and the CaTSSITTR at Ivanpah Playa and in S. Dakota is shown in Figure 11.



Figure 11. Surface BRF measurements. (**a**) UArizona personnel operating a portable spectroradiometer; (**b**) NASA GSFC personnel operating the CaTSSITTR; (**c**) SDSU personnel operating a portable spectroradiometer at a vegetated site. Ivanpah photos courtesy of L. Thome.

2.3.5. TOA Spectral Radiance Determination

The final data processing step in the reflectance-based approach is the determination of the TOA quantities of interest using the radiative transfer code [31]. The spatially averaged surface BRF data are used with the atmospheric measurements as input into the radiative transfer code, and the result is the band-averaged TOA spectral radiance and reflectance, which are then compared to the Landsat 9 OLI data products. An example of the TOA spectral radiance determined at one of the SDSU vegetated sites for a Landsat 9 OLI collection on 18 August 2023 is shown in Figure 12. A general summary of the reflectance-based approach processing methodology is shown in Figure 13.



Figure 12. TOA spectral radiance at an SDSU vegetated site on 18 August 2023 at 17:11 UTC for the Landsat 9 OLI viewing conditions.



Figure 13. A summary of the reflectance-based processing methodology.

Band averaged TOA spectral radiance and TOA reflectance

3. Data

3.1. Landsat 9 OLI Imagery

This work uses the Landsat Collection 2 Level-1TP (L1TP) data for comparison with the ground measurements to evaluate the absolute radiometric calibration of the OLI sensor. The L1TP imagery is produced by the Landsat Product Generation System (LPGS), where each image is radiometrically calibrated and orthorectified using ground control points (GCPs) and digital elevation models (DEMs). The imagery is freely available to registered users on the USGS Earth Explorer web portal. As with the Landsat 8 OLI imagery, there are effectively two data radiometry products of interest in each L1TP image as follows: the TOA spectral radiance and the TOA reflectance. The Landsat data archive was restructured into collections in 2016 to ensure that the radiometry and geometry of the Level-1 data products are consistent over the entire data record [108]. Collection 1 included Landsat 1–5, 7, and 8, and it was available until December 2022. It has been superseded with the release of Collection 2, which adds Landsat 9 to the data archive. Updates to Collection 2 include radiometric and geometric improvements, as well as the inclusion of Level-2 surface reflectance and surface temperature data products [109]. The Collection 2 L1TP imagery is converted from a digital number (DN) to either TOA spectral radiance or TOA reflectance using radiometric scaling coefficients that are included in the metadata file accompanying the imagery. It should be noted that the TOA spectral radiance and TOA reflectance products follow separate radiometric calibration traceability chains, so unlike previous Landsat sensors, there is no official exoatmospheric solar irradiance model to convert from one to the other.

Preliminary L1 imagery was made available to the ground calibration teams after the Landsat 9 commissioning phase to monitor any post-launch radiometric issues with the OLI. The calibration parameter files (CPFs) have since been updated, and the work presented here uses the most current CPFs available. As mentioned, the USGS also produces a Level-2 surface reflectance data product (L2 SR) using the Land Surface Reflectance Code (LaSRC), but the scope of this work is limited to radiometric validation.

3.2. RadCaTS

A summary of the satellite ephemeris for Landsat 9 at Railroad Valley is shown in Table 2, which describes the collection area of the ground site, the typical overpass time, and the satellite view geometry. There were 38 Landsat 9 daytime overpasses at the RadCaTS during the period from November 2021 to June 2023, with 12 being considered successful. The typical reasons for an unsuccessful collection are clouds at the site or water/snow on the test site surface. The current quality assurance (QA) criteria for an overpass at the RadCaTS is similar to what was developed for the RadCalNet [110]. The QA criteria are founded on a well-understood uncertainty budget and include the 550 nm AOD as well as the average surface reflectance for each of the eight GVR channels to account for surface nonuniformities.

Table 2. A summary of the typical overpass time, satellite view zenith angle (VZA), and view azimuth angle (VAA) conditions for Landsat 9 at the RadCaTS. Note that the VAA defined here is measured from the ground site to the satellite.

RadCaTS (Railroad Valley, NV, USA)				
Group	UArizona			
Number of Collects	12			
Collection Area (m)	1000 imes 1000			
Time (UTC)	18:21			
VZA (degrees)	0.5			
VAA (degrees)	103.0			

3.3. Reflectance-Based Approach (SDSU, UArizona, and GSFC)

A summary of the satellite ephemeris for the Landsat 9 OLI at the SDSU sites near Brookings and the joint UArizona–GSFC field campaign at Ivanpah Playa is shown in Table 3. It should be noted that the UArizona and GSFC work at Ivanpah occurred during the tandem flight period of Landsat 8 and Landsat 9, and data are not collected on a routine basis at Ivanpah following this field campaign. The Ivanpah collection was considered successful, so the results are included in the overall analysis. SDSU had three successful Landsat 9 collects in June, July, and August 2022, and they are included in the overall results as well. The reason for an unsuccessful collection at SDSU is typically due to poor atmospheric conditions and water or snow on the site.

Table 3. An overview of the overpass time, satellite view zenith angle (VZA), and view azimuth angle (VAA) conditions for Landsat 9 at the sites used by SDSU, UArizona, and GSFC. The VAA is defined as measured from the ground site to the satellite.

Ground Site	Brookings, SD, USA	Ivanpah, CA, USA
Group	SDSU	UArizona & GSFC
Number of Collects	3	1 each
	150 imes250	
Collection Area(s) (m)	120 imes180	120×300
	120 imes 180	
Time (UTC)	17:11	18:20
VZA (degrees)	0.6	3.8
VAA (degrees)	104.2	283.5

4. Results

4.1. RadCaTS Results

There were 12 successful collects at RadCaTS in the 19 months following the commissioning phase of Landsat 9. The comparison between the OLI and RadCaTS for both the TOA spectral radiance and TOA reflectance data products is shown in Figure 14 as the ratio of the OLI values to those determined using the RadCaTS. There was a higher amount of snow and precipitation than normal at Railroad Valley from January 2023 to April 2023, which is why there is a lower number of successful collects than would be anticipated. Efflorescent salts typically become present while the playa is drying, which can lead to surface uniformity issues when using point measurements, such as those with a finite number of GVRs.



Figure 14. A summary of the Landsat 9 OLI radiometric calibration results using 12 comparisons with the RadCaTS (November 2021 to June 2023). The results are shown as the ratio of the TOA quantities for each data product (e.g., spectral radiance or reflectance) to those determined using the RadCaTS.

The RadCaTS results for each of the two data products agree to within the uncertainties of the methodology, which is summarized here as $\pm 4\%$ (k = 1). One noticeable result in Figure 14 is that of the slight offset between the results of the two L1TP imagery products. This bias is highest in the Coastal Aerosol band (band 1) and the SWIR 2 band (band 7), which are those most affected by atmospheric effects and the low SNR of the ground data. This pattern has also been observed in previous Landsat 8 work, but it should be noted that the offset is within the combined uncertainties of the methodologies.

4.2. Reflectance-Based Results at SDSU

There were three successful ground-based collects for Landsat 9 during the period from November 2021 to June 2023 using the SDSU vegetated sites in Brookings. The average values are shown in Figure 15, where the format is identical to Figure 14. The uncertainty bars are based on the current uncertainty estimate for the reflectance-based approach of $\pm 2.5\%$ (k = 1) [17]. In OLI bands 1 and 2 (coastal aerosol and blue), there is a higher bias between the OLI and the field measurements, and this is mainly due to the very low surface reflectance of vegetation in these two bands coupled with atmospheric effects that occur for a surface reflectance of <0.2 in these spectral regions. An example of the typical surface BRF as a function of wavelength for Railroad Valley, Ivanpah Playa, and an SDSU vegetated site is shown in Figure 16.



Figure 15. A summary of the SDSU radiometric calibration results for the Landsat 9 OLI using three comparisons with ground measurements using the reflectance-based approach (June 2022 to August 2022). The results are shown as the ratio of the TOA quantities for each data product (e.g., spectral radiance or reflectance) to the in situ field results.



Figure 16. An example of the typical surface BRF at Railroad Valley (RRV), Ivanpah Playa, and S. Dakota State University (SDSU) as a function of wavelength.

4.3. Reflectance-Based Results for UArizona and NASA GSFC at Ivanpah Playa

The joint UArizona–GSFC field deployment to Ivanpah Playa was successful, with clear skies and minimal wind, which can lead to airborne dust. Each team collected data within the predefined ROI using a portable spectroradiometer (UArizona) and CaTSSITTR (GSFC). Landsat 9 OLI results from the joint UArizona–GSFC collection at Ivanpah Playa during the tandem flight phase with Landsat 8 are shown in Figure 17. The uncertainty bars reflect the current uncertainty estimate for the reflectance-based approach of $\pm 2.5\%$ (k = 1). The results from both teams agree to within the uncertainties of the reflectance-based approach.





4.4. Summary of Combined Landsat 9 OLI Results

A summary of the Landsat 9 OLI results using all data from the three field teams is shown in Figure 18 for the period from November 2021 to June 2023. The results comprise a total of 17 ground-based collects using the RadCaTS and the reflectance-based approach, and the uncertainty bars in this case are the standard deviation (k = 1) of the field results. The results show excellent agreement with both OLI TOA products and are well within the specified radiometric uncertainties of the OLI.



Figure 18. A summary of the radiometric calibration results for the Landsat 9 OLI using all available ground-based data, which include the RadCaTS and the reflectance-based results from three groups (November 2021–June 2023). The results are shown as the ratio of the TOA quantities for each data product (e.g., spectral radiance or reflectance) to the ground-based data, and the uncertainty bars for each band are the standard deviation (k = 1) of the field measurements. The same results in tabular form are presented in Table 4.

A summary of the temporal results for TOA spectral radiance is shown in Figure 19, which includes all data from each field team. The results do not show a statistically significant trend, but any such potential trend will become apparent once more temporal data are acquired.



Figure 19. The TOA spectral radiance temporal results for the Landsat 9 OLI, comprising all data from each ground team during the period from November 2021 to June 2023.

Ratio of TOA Quantities to Ground Measurements—All Field Data					
OLI Band (Center Wavelength)	TOA Spectral Radiance (OLI/ground)	TOA Reflectance (OLI/ground)			
1. (443 nm)	1.031 ± 0.051	0.989 ± 0.046			
2. (483 nm)	1.014 ± 0.046	0.994 ± 0.041			
3. (561 nm)	1.003 ± 0.043	0.999 ± 0.040			
4. (655 nm)	1.018 ± 0.046	1.020 ± 0.045			
5. (865 nm)	1.024 ± 0.045	1.014 ± 0.043			
6. (1609 nm)	1.008 ± 0.043	1.021 ± 0.045			
7. (2201 nm)	0.979 ± 0.040	1.002 ± 0.040			
8. (pan, 592 nm)	1.011 ± 0.043	0.995 ± 0.041			

Table 4. The radiometric calibration results for Landsat 9 OLI using all ground data (from November 2021 to June 2023).

One interesting result from the tandem flight phase with Landsat 8 in November 2021 is observed in both the in situ data collected at Ivanpah by two separate ground teams (Figure 17) and at Railroad Valley using the RadCaTS (Figure 19). The Landsat 9 near-nadir overpass at each of these sites on 13 November 2021 is separated by 45 s, and the data from all three collects show similar results in the ratio of OLI TOA values to the ground measurements. This result is encouraging, as the two systems have a different uncertainty traceability path, and they produce results within the combined uncertainty of each methodology. The results from this single day have a smaller ratio of TOA spectral radiance (OLI/ground) than most of the other dates in this work, but the fact that it occurs at two separate sites with separate traceability paths does increase confidence in the result. The results of this work do not show a significant trend in the radiometric calibration of Landsat 9 OLI, but new data are being routinely collected by the RadCaTS and SDSU for future analysis.

5. Uncertainty Analysis

A detailed uncertainty analysis of both the reflectance-based approach and the Rad-CaTS has been previously documented, and only a summary is presented here [17,110,111]. The uncertainty analysis for the RadCaTS was completed as part of the requirements for inclusion as an official RadCalNet site, and the uncertainty of the reflectance-based approach was most recently analyzed for work supporting the launch of Landsat 8. The current non-spectral estimate of the RadCaTS uncertainty is approximately $\pm 5\%$ from 400 nm to 2400 nm, and the estimate for the reflectance-based approach is approximately $\pm 2.6\%$ in the middle of the visible spectral region.

6. Conclusions

After an extensive ground-based field measurement campaign by various government and university groups during the first nineteen months of operation, it has been confirmed that the Landsat 9 OLI instrument is performing exceptionally well and has maintained its stability throughout this period. The ground-based data were compared to the L1TP imagery available from the USGS through Earth Explorer. The results showed that, on average, the OLI instrument aligns with the design specifications for TOA spectral radiance (\pm 5%) and TOA spectral reflectance (\pm 3%) in all multispectral bands as well as the panchromatic band (band 8).

The temporal analysis did not reveal any significant degradation pattern in the OLI instrument over time. Furthermore, the results obtained from the automated RadCaTS facility at Railroad Valley supported the findings of the ground personnel, with even slightly better agreement with the OLI than the in situ measurements conducted by on-site personnel. However, it is worth noting that the number of RadCaTS data sets was lower than expected throughout the year, primarily due to the unusually high frequency of poor weather conditions experienced during that period.

The radiometric calibration of the Landsat 9 OLI will continue to be measured and monitored throughout its operational life, employing reliable and established techniques that are currently in use by a wide array of calibration teams. These techniques include ground-based in situ measurements using both automated systems and on-site personnel, observations over invariant desert sites, lunar observations, and cross-comparison with other well-calibrated sensors. By utilizing these methods, the temporal accuracy and precision of Landsat 9 OLI data will remain well understood, thereby ensuring the continued production of high-quality and reliable Earth observation data for a wide range of applications.

Author Contributions: All the authors were involved in various aspects of the data collection and/or analysis during the period of this work. J.S.C.-M. and N.J.A. were involved in the in situ collection and analysis of field data, which included simultaneous field work with K.J.T. and B.N.W., who provided the NASA GSFC results. C.T.P. and L.M.L. were involved in on-site data collection at the SDSU calibration site, including supervision, and they also assumed responsibility for the subsequent processing and analysis of field data. All authors have read and agreed to the published version of the manuscript.

Funding: The UArizona work was supported by the USGS cooperative agreement G20AC00003, as well as NASA grants 80NSSC21K1749 and 80NSSC20K1872. The SDSU work was supported by USGS EROS grant G18AS00001.

Data Availability Statement: Data from the RadCaTS are freely available to registered users on the Radiometric Calibration Network (RadCalNet) web portal: www.radcalnet.org (accessed on 1 January 2024). Landsat 8 and 9 OLI Collection 2 data are courtesy of the US Geological Survey DOI Object Identifier (DOI) number: /10.5066/P975CC9B.

Acknowledgments: The authors would like the thank the U.S. Bureau of Land Management (BLM), Tonopah, Nevada, office for assistance and permission for the use of Railroad Valley and the BLM, Needles, California, office for their assistance in using Ivanpah Playa. We would also like to thank NASA AERONET for processing the Cimel sun photometer data, the NOAA Global Monitoring Laboratory (GML) for the CO_2 data, and the OMI and OMPS teams for the ozone data.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

The surface BRF, ρ , for each of the GVR channels is determined using the following equation:

$$\rho = \frac{\pi C_{GVR} \left(V_{GVR} - V_{GVR \ dark} \right)}{\frac{E_0}{d^2} \tau_A \cos\theta + E_{sky}}$$

where C_{GVR} is the radiometric calibration coefficient of the GVR (W m⁻² sr⁻¹ µm⁻¹) V⁻¹, V_{GVR} is the output voltage (V), $V_{GVR \, dark}$ is the dark voltage (V), E_0 is the exoatmospheric spectral solar irradiance when the Earth–Sun distance is one astronomical unit (AU) (W m⁻² µm⁻¹), d is the Earth–Sun distance (AU), τ_A is the direct solar beam transmission (unitless), θ is the solar zenith angle, and E_{sky} is the diffuse spectral sky irradiance (W m⁻² µm⁻¹).

References

- Loveland, T.R.; Anderson, M.C.; Huntington, J.L.; Irons, J.R.; Johnson, D.M.; Rocchio, L.E.; Woodcock, C.E.; Wulder, M.A. Seeing Our Planet Anew: Fifty Years of Landsat. *Photogramm. Eng. Remote Sens.* 2022, *88*, 429–436. [CrossRef]
- Masek, J.G.; Wulder, M.A.; Markham, B.; McCorkel, J.; Crawford, C.J.; Storey, J.; Jenstrom, D.T. Landsat 9: Empowering open science and applications through continuity. *Remote Senssing Environ.* 2020, 248, 111968. [CrossRef]
- 3. Wulder, M.A.; Roy, D.P.; Radeloff, V.C.; Loveland, T.R.; Anderson, M.C.; Johnson, D.M.; Healey, S.; Zhu, Z.; Scambos, T.A.; Pahlevan, N. Fifty years of Landsat science and impacts. *Remote Sens. Environ.* **2022**, *280*, 113195. [CrossRef]
- Saralioglu, E.; Vatandaslar, C. Land use/land cover classification with Landsat-8 and Landsat-9 satellite images: A comparative analysis between forest- and agriculture-dominated landscapes using different machine learning methods. *Acta Geod. Geophys.* 2022, 57, 695–716. [CrossRef]
- 5. Knight, E.J.; Kvaran, G. Landsat-8 Operational Land Imager Design, Characterization and Performance. *Remote Sens.* 2014, *6*, 10286–10305. [CrossRef]

- Levy, R.; Markham, B.L. Landsat 9 Operational Land Imager2 (OLI2) diffuser panel response lab predictions vs. pre-launch measurements. In Proceedings of the Earth Observing Systems XXV, Online, 17 September 2020; pp. 103–117.
- Micijevic, E.; Haque, M.O.; Scaramuzza, P.; Storey, J.; Anderson, C.; Markham, B. Landsat 9 pre-launch sensor characterization and comparison with Landsat 8 results. In Proceedings of the Sensors, Systems, and Next-Generation Satellites XXIII, Strasbourg, France, 10 October 2019; pp. 289–300.
- Markham, B.; Barsi, J.; Donley, E.; Efremova, B.; Hair, J.; Jenstrom, D.; Kaita, E.; Knight, E.; Kvaran, G.; McCorkel, J.; et al. Landsat 9: Mission Status and Prelaunch Instrument Performance Characterization and Calibration. In Proceedings of the IGARSS 2019—2019 IEEE International Geoscience and Remote Sensing Symposium, Yokohama, Japan, 28 July–2 August 2019; pp. 5788–5791.
- Barsi, J.A.; Markham, B.L.; McCorkel, J.; McAndrew, B.; Donley, E.; Morland, E.; Pharr, J.; Rodriguez, M.; Shuman, T.; Sushkov, A. The operational land Imager-2: Prelaunch spectral characterization. In Proceedings of the Earth Observing Systems XXIV, San Diego, CA, USA, 9 September 2019; pp. 35–45.
- Malone, K.J.; Schrein, R.J.; Bradley, M.S.; Irwin, R.; Berdanier, B.; Donley, E. Landsat 9 OLI 2 focal plane subsystem: Design, performance, and status. In Proceedings of the Earth Observing Systems XXII, San Diego, CA, USA, 5 September 2017; pp. 42–62.
- Bouvet, M.; Thome, K.; Berthelot, B.; Bialek, A.; Czapla-Myers, J.; Fox, N.P.; Goryl, P.; Henry, P.; Ma, L.; Marcq, S.; et al. RadCalNet: A Radiometric Calibration Network for Earth Observing Imagers Operating in the Visible to Shortwave Infrared Spectral Range. *Remote Sens.* 2019, 11, 2401. [CrossRef]
- 12. Slater, P.N.; Biggar, S.F.; Thome, K.J.; Gellman, D.I.; Spyak, P.R. Vicarious Radiometric Calibrations of EOS Sensors. J. Atmos. Ocean. Technol. 1996, 13, 349–359. [CrossRef]
- 13. Teillet, P.M.; Slater, P.N.; Ding, Y.; Santer, R.P.; Jackson, R.D.; Moran, M.S. Three Methods for the Absolute Calibration of the NOAA AVHRR Sensors In-Flight. *Remote Sens. Environ.* **1990**, *31*, 105–120. [CrossRef]
- 14. Slater, P.N.; Biggar, S.F.; Holm, R.G.; Jackson, R.D.; Mao, Y.; Moran, M.S.; Palmer, J.M.; Yuan, B. Reflectance- and radiance-based methods for the in-flight absolute calibration of multispectral sensors. *Remote Sens. Environ.* **1987**, *22*, 11–37. [CrossRef]
- 15. Wenny, B.; Thome, K.; Czapla-Myers, J. Evaluation of vicarious calibration for airborne sensors using RadCalnet. In Proceedings of the Sensors, Systems, and Next-Generation Satellites XXIV, Online, 2 October 2020.
- Lau, I.C.; Ong, C.C.H.; Thome, K.J.; Wenny, B.; Mueller, A.; Heiden, U.; Czapla-Myers, J.; Biggar, S.; Anderson, N.; McGonigle, L.; et al. Intercomparison of Field Methods for Acquiring Ground Reflectance at Railroad Valley Playa for Spectral Calibration of Satellite Data. In Proceedings of the IGARSS 2018—2018 IEEE International Geoscience and Remote Sensing Symposium, Valencia, Spain, 22–27 July 2018; pp. 186–188.
- 17. Czapla-Myers, J.; McCorkel, J.; Anderson, N.; Thome, K.; Biggar, S.; Helder, D.; Aaron, D.; Leigh, L.; Mishra, N. The Ground-Based Absolute Radiometric Calibration of Landsat 8 OLI. *Remote Sens.* **2015**, *7*, 600–626. [CrossRef]
- 18. Thome, K.J. Absolute radiometric calibration of Landsat 7 ETM+ using the reflectance-based method. *Remote Sens. Environ.* **2001**, 78, 27–38. [CrossRef]
- Shrestha, M.; Hasan, N.; Leigh, L.; Helder, D. Derivation of Hyperspectral Profile of Extended Pseudo Invariant Calibration Sites (EPICS) for Use in Sensor Calibration. *Remote Sens.* 2019, 11, 2279. [CrossRef]
- Russell, B.; Holt, J.; Durell, C.; Arnold, W.; Conran, D.; Schiller, S. The Flare: Network: Autonomous, On-Demand Spatial and Radiometric Calibration and Validation for Imaging Spectroscopy. In Proceedings of the 2021 IEEE International Geoscience and Remote Sensing Symposium IGARSS, Brussels, Belgium, 11–16 July 2021; pp. 1615–1618.
- 21. Holt, J.; Durell, C.; Russell, B.; Conran, D.; Arnold, W.; Schiller, S. FLARE network performance: Automated on-demand calibration for space, airborne and UAV assets. In Proceedings of the SPIE Defense + Commercial Sensing, Online, 12 April 2021.
- 22. Slater, P.N. Radiometric considerations in remote sensing. *Proc. IEEE* **1985**, 73, 997–1011. [CrossRef]
- Slater, P.N. The Importance and Atteinment of Accurate Absolute Radiometric Calibration. *Remote Sens. Crit. Rev. Technol.* 1984, 475, 34–41.
- Kastner, C.; Slater, P. In-Flight Radiometric Calibration of Advanced Remote Sensing Systems. In Proceedings of the 26th Annual Technical Symposium, Field Measurement and Calibration Using Electro-Optical Equipment, San Diego, CA, USA, 23 June 1983; pp. 158–165.
- 25. Schaepman-Strub, G.; Schaepman, M.E.; Painter, T.H.; Dangel, S.; Martonchik, J.V. Reflectance quantities in optical remote sensing—Definitions and case studies. *Remote Sens. Environ.* **2006**, *103*, 27–42. [CrossRef]
- 26. Ehsani, A.R.; Reagan, J.A.; Erxleben, W.H. Design and Performance Analysis of an Automated 10-Channel Solar Radiometer Instrument. J. Atmos. Ocean. Technol. 1998, 15, 697–707. [CrossRef]
- 27. Thome, K.J.; Smith, M.W.; Palmer, J.M.; Reagan, J.A. Three-channel solar radiometer for the determination of atmospheric columnar water vapor. *Appl. Opt.* **1994**, *33*, 5811–5819. [CrossRef]
- 28. Reagan, J.A.; Thome, K.J.; Herman, B.M. A Simple Instrument and Technique for Measuring Columnar Water Vapor via Near-IR Differential Solar Transmission Measurments. *IEEE Trans. Geosci. Remote Sens.* **1992**, *30*, 825–831. [CrossRef]
- Barreto, Á.; Cuevas, E.; Granados-Muñoz, M.J.; Alados-Arboledas, L.; Romero, P.M.; Gröbner, J.; Kouremeti, N.; Almansa, A.F.; Stone, T.; Toledano, C.; et al. The new sun-sky-lunar Cimel CE318-T multiband photometer—A comprehensive performance evaluation. *Atmos. Meas. Tech.* 2016, 9, 631–654. [CrossRef]

- Holben, B.N.; Eck, T.F.; Slutsker, I.; Tanré, D.; Buis, J.P.; Setzer, A.; Vermote, E.; Reagan, J.A.; Kaufman, Y.J.; Nakajima, T.; et al. AERONET—A Federated Instrument Network and Data Archive for Aerosol Characterization. *Remote Sens. Environ.* 1998, 66, 1–16. [CrossRef]
- Berk, A.; Conforti, P.; Kennett, R.; Perkins, T.; Hawes, F.; van den Bosch, J. MODTRAN6: A major upgrade of the MODTRAN radiative transfer code. In Proceedings of the 2014 6th Workshop on Hyperspectral Image and Signal Processing: Evolution in Remote Sensing (WHISPERS), Lausanne, Switzerland, 24–27 June 2014.
- 32. Czapla-Myers, J.; Thome, K.; Biggar, S. Unmanned vicarious calibration for large-footprint sensors. In Proceedings of the Earth Observing Systems X, San Diego, CA, USA, 22 August 2005; Volume 5882, pp. 416–425.
- 33. Thome, K.J.; Czapla-Myers, J.S.; Biggar, S.F. Ground-monitor radiometer system for vicarious calibration. In Proceedings of the Imaging Spectrometry X, Denver, CO, USA, 15 October 2004; pp. 223–232.
- Czapla-Myers, J.S.; Thome, K.J.; Biggar, S.F. Optical sensor package for multiangle measurements of surface reflectance. In Proceedings of the Imaging Spectrometry VII, San Diego, CA, USA, 17 January 2002; pp. 326–333.
- Anderson, N.J.; Czapla-Myers, J.S. Ground viewing radiometer characterization, implementation and calibration applications: A summary after two years of field deployment. In Proceedings of the SPIE Optical Engineering+ Applications, San Diego, CA, USA, 23 September 2013; Volume 8866, pp. 185–194.
- Anderson, N.; Czapla-Myers, J.; Leisso, N.; Biggar, S.; Burkhart, C.; Kingston, R.; Thome, K. Design and calibration of field deployable ground-viewing radiometers. *Appl. Opt.* 2013, 52, 231–240. [CrossRef] [PubMed]
- Singh, R.; Czapla-Myers, J.; Anderson, N. Ground viewing radiometer equipped with autonomous linear motion: Two year field deployment summary and analysis. In Proceedings of the SPIE Optical Engineering + Applications, San Diego, CA, USA, 30 September 2022.
- Czapla-Myers, J.; McCorkel, J.; Anderson, N.; Biggar, S. Earth-observing satellite intercomparison using the Radiometric Calibration Test Site at Railroad Valley. J. Appl. Remote Sens. 2017, 12, 012004. [CrossRef]
- 39. Markham, B.; Barsi, J.; Kvaran, G.; Ong, L.; Kaita, E.; Biggar, S.; Czapla-Myers, J.; Mishra, N.; Helder, D. Landsat-8 Operational Land Imager Radiometric Calibration and Stability. *Remote Sens.* **2014**, *6*, 12275–12308. [CrossRef]
- Czapla-Myers, J.S.; Anderson, N.J.; Biggar, S.F. Early ground-based vicarious calibration results for Landsat 8 OLI. In Proceedings
 of the Earth Observing Systems XVIII, San Diego, CA, USA, 23 September 2013; Volume 8866, pp. 205–214.
- 41. Czapla-Myers, J.S.; Thome, K.; Biggar, S.; Anderson, N.J. The absolute radiometric calibration of Terra imaging sensors: MODIS, MISR, and ASTER. In Proceedings of the Earth Observing Systems XIX, San Diego, CA, USA, 2 October 2014.
- Yamamoto, H.; Czapla-Myers, J.; Tsuchida, S. Validation of Aster VNIR Radiometric Performance Using the Reflectance-Based Vicarious Calibration Experiments and RadCaTS Data. In Proceedings of the IGARSS 2022—2022 IEEE International Geoscience and Remote Sensing Symposium, Kuala Lumpur, Malaysia, 17–22 July 2022; pp. 4316–4319.
- Tsuchida, S.; Yamamoto, H.; Kouyama, T.; Obata, K.; Sakuma, F.; Tachikawa, T.; Kamei, A.; Arai, K.; Czapla-Myers, J.S.; Biggar, S.F.; et al. Radiometric Degradation Curves for the ASTER VNIR Processing Using Vicarious and Lunar Calibrations. *Remote Sens.* 2020, 12, 427. [CrossRef]
- 44. Tahersima, M.H.; Wenny, B.N.; Voskanian, N.; Thome, K. Intercomparison of Landsat and Joint Polar Satellite System using RadCalNet. In Proceedings of the Earth Observing Systems XXVII, San Diego, CA, USA, 30 September 2022; pp. 352–357.
- 45. Czapla-Myers, J.S.; Anderson, N.J. Intercomparison of the GOES-16 and -17 Advanced Baseline Imager with low-Earth orbit sensors. In Proceedings of the Earth Observing Systems XXIV, San Diego, CA, USA, 9 September 2019.
- 46. Czapla-Myers, J.; Anderson, N. Post-launch radiometric validation of the GOES-16 Advanced Baseline Imager (ABI). In Proceedings of the SPIE Remote Sensing, Berlin, Germany, 25 September 2018.
- 47. Wenny, B.; Thome, K.; Czapla-Myers, J. Evaluation of vicarious calibration for airborne sensors using RadCalNet. J. Appl. Remote Sens. 2021, 15, 034501. [CrossRef]
- 48. Bruegge, C.J.; Arnold, G.T.; Czapla-Myers, J.; Dominguez, R.A.; Helmlinger, M.C.; Thompson, D.R.; Van den Bosch, J.; Wenny, B. Vicarious Calibration of eMAS, AirMSPI, and AVIRIS Sensors during FIREX-AQ. *TGARS* **2021**, *59*, 10286–10297. [CrossRef]
- Martins, J.V.; Fernandez-Borda, R.; McBride, B.; Remer, L.; Barbosa, H.M.J. The Harp Hype Ran Gular Imaging Polarimeter and the Need for Small Satellite Payloads with High Science Payoff for Earth Science Remote Sensing. In Proceedings of the IGARSS 2018—2018 IEEE International Geoscience and Remote Sensing Symposium, Valencia, Spain, 22–27 July 2018; pp. 6304–6307.
- 50. Houborg, R.; McCabe, M.F. A Cubesat enabled Spatio-Temporal Enhancement Method (CESTEM) utilizing Planet, Landsat and MODIS data. *Remote Sens. Environ.* 2018, 209, 211–226. [CrossRef]
- 51. Wilson, N.; Greenberg, J.; Jumpasut, A.; Collison, A.; Weichelt, H. *Absolute Radiometric Calibration of Planet Dove Satellites, Flocks 2p* & 2e; Planet: San Francisco, CA, USA, 2017.
- Pack, D.; Ardila, D.; Herman, E.; Rowen, D.; Welle, R.; Wiktorowicz, S.; Hattersley, B. Two Aerospace Corporation CubeSat Remote Sensing Imagers: CUMULOS and R3. In Proceedings of the AIAA/USU Small Satellite Conference, Logan, UT, USA, 8 August 2017.
- 53. Marchant, A.B. Design and demonstration of a CubeSat-scale spatial heterodyne imaging spectrometer. In Proceedings of the SPIE Optical Engineering + Applications, San Diego, CA, USA, 19 September 2016.
- 54. Verstraete, M.M.; Diner, D.J.; Bézy, J.-L. Planning for a spaceborne Earth Observation mission: From user expectations to measurement requirements. *Environ. Sci. Policy* **2015**, *54*, 419–427. [CrossRef]

- 55. Belward, A.S.; Skøien, J.O. Who launched what, when and why; trends in global land-cover observation capacity from civilian earth observation satellites. *ISPRS J. Photogramm. Remote Sens.* **2015**, *103*, 115–128. [CrossRef]
- 56. Puschell, J.J.; Stanton, E. CubeSat modules for multispectral environmental imaging from polar orbit. In Proceedings of the SPIE Optical Engineering + Applications, San Diego, CA, USA, 23 October 2012.
- Naughton, D.; Brunn, A.; Czapla-Myers, J.; Douglass, S.; Thiele, M.; Weichelt, H.; Oxfort, M. Absolute radiometric calibration of the RapidEye multispectral imager using the reflectance-based vicarious calibration method. *J. Appl. Remote Sens.* 2011, *5*, 053544. [CrossRef]
- Brunn, A.; Naughton, D.; Weichelt, H.; Douglass, S.; Thiele, M.; Oxfort, M.; Beckett, K. The calibration procedure of the multispectral imaging instruments on board the RapidEye remote sensing Satellites. In Proceedings of the International Calibration and Orientation Workshop, EuroCow, Castelldefels, Spain, 10–12 February 2010; pp. 10–12.
- Ong, C.; Caccetta, M.; Lau, I.C.; Ong, L.; Middleton, E. Compositional characterisation of the pinnacles vicarious calibration site. In Proceedings of the 2017 IEEE International Geoscience and Remote Sensing Symposium (IGARSS), Fort Worth, TX, USA, 23–28 July 2017; pp. 3059–3062.
- Pato, M.; Bachmann, M.; de los Reyes, R.; Alonso, K.; Baur, S.; Gerasch, B.; Holzwarth, S.; Langheinrich, M.; Marshall, D.; Schneider, M.; et al. First year EnMAP radiometric performance based on scenes over RadCalNet and PICS sites. In Proceedings of the Optica Sensing Congress 2023 (AIS, FTS, HISE, Sensors, ES), Munich, Germany, 30 July 2023; p. HM2C.5.
- 61. Angal, A.; Bruegge, C.; Xiong, X.; Wu, A. Intercalibration of the reflective solar bands of MODIS and MISR instruments on the Terra platform. *J. Appl. Remote Sens.* **2022**, *16*, 027501. [CrossRef]
- 62. Angal, A.; Xiong, X.; Thome, K.; Wenny, B.N. Cross-Calibration of Terra and Aqua MODIS Using RadCalNet. *IEEE Geosci. Remote Sens. Lett.* **2021**, *18*, 188–192. [CrossRef]
- 63. Gao, C. An Approach for Evaluating Multi-Site Radiometry Calibration of Sentinel-2B/MSI using RadCalNet Sites. *IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens.* **2021**, *14*, 8473–8483. [CrossRef]
- 64. Shrestha, M.; Helder, D.; Christopherson, J. DLR Earth Sensing Imaging Spectrometer (DESIS) Level 1 Product Evaluation Using RadCalNet Measurements. *Remote Sens.* 2021, *13*, 2420. [CrossRef]
- Wolters, E.; Toté, C.; Sterckx, S.; Adriaensen, S.; Henocq, C.; Bruniquel, J.; Scifoni, S.; Dransfeld, S. iCOR Atmospheric Correction on Sentinel-3/OLCI Over Land: Intercomparison with AERONET, RadCalNet, and SYN Level-2. *Remote Sens.* 2021, 13, 654. [CrossRef]
- 66. Kim, K.; Lee, K. A Validation Experiment of the Reflectance Products of KOMPSAT-3A Based on RadCalNet Data and Its Applicability to Vegetation Indexing. *Remote Sens.* 2020, 12, 3971. [CrossRef]
- 67. Jing, X.; Leigh, L.; Teixeira Pinto, C.; Helder, D. Evaluation of RadCalNet Output Data Using Landsat 7, Landsat 8, Sentinel 2A, and Sentinel 2B Sensors. *Remote Sens.* 2019, *11*, 541. [CrossRef]
- Sterckx, S.; Wolters, E. Radiometric Top-of-Atmosphere Reflectance Consistency Assessment for Landsat 8/OLI, Sentinel-2/MSI, PROBA-V, and DEIMOS-1 over Libya-4 and RadCalNet Calibration Sites. *Remote Sens.* 2019, 11, 2253. [CrossRef]
- Banks, A.C.; Hunt, S.E.; Gorroño, J.; Scanlon, T.; Woolliams, E.R.; Fox, N.P. A comparison of validation and vicarious calibration of high and medium resolution satellite-borne sensors using RadCalNet. In Proceedings of the SPIE Remote Sensing, Warsaw, Poland, 29 September 2017; pp. 246–254.
- de los Reyes, R.; Alonso, K.; Bachmann, M.; Carmona, E.; Langheinrich, M.; Müller, R.; Pflug, B.; Richter, R. The desis l2a processor and validation of l2a products using aeronet and radcalnet data. *Int. Arch. Photogramm. Remote Sens. Spat. Inf. Sci.* 2022, 46, 9–12. [CrossRef]
- 71. Mims, F.M., III. An inexpensive and stable LED Sun photometer for measuring the water vapor column over South Texas from 1990 to 2001. *Geophys. Res. Lett.* 2002, 29, 20-1–20-4. [CrossRef]
- 72. Brooks, D.R.; Mims III, F.M. Development of an inexpensive handheld LED-based Sun photometer for the GLOBE program. J. *Geophys. Res.* 2001, 106, 4733–4740. [CrossRef]
- 73. Mims, F.M., III. An International Haze-Monitoring Network for Students. Bull. Am. Meteorol. Soc. 1999, 80, 1421–1431. [CrossRef]
- 74. Mims, F.M., III. Sun photometer with light-emitting diodes as spectrally selective detectors. *Appl. Opt.* **1992**, *31*, 6965–6967. [CrossRef]
- 75. Czapla-Myers, J.S.; Thome, K.J.; Biggar, S.F. Design, calibration, and characterization of a field radiometer using light-emitting diodes as detectors. *Appl. Opt.* **2008**, *47*, 6753–6762. [CrossRef]
- Czapla-Myers, J.S.; Thome, K.J.; Cocilovo, B.R.; McCorkel, J.T.; Buchanan, J.H. Temporal, spectral, and spatial study of the automated vicarious calibration test site at Railroad Valley, Nevada. In Proceedings of the Earth Observing Systems XIII, San Diego, CA, USA, 20 August 2008; Volume 7081, pp. 161–169.
- Czapla-Myers, J.S.; Thome, K.J.; Buchanan, J.H. Implication of spatial uniformity on vicarious calibration using automated test sites. In Proceedings of the Earth Observing Systems XII, San Diego, CA, USA, 27 September 2007; Volume 6677, pp. 327–336.
- Scott, K.P.; Thome, K.J.; Brownlee, M.R. Evaluation of Railroad Valley Playa for use in vicarious calibration. In Proceedings of the SPIE, Multispectral Imaging for Terrestrial Applications, Denver, CO, USA, 4 November 1996; pp. 158–166.
- Brown, S.W.; Johnson, B.C.; Yoon, H.W.; Butler, J.J.; Barnes, R.A.; Biggar, S.F.; Spyak, P.R.; Thome, K.J.; Zalewski, E.F.; Helmlinger, M.; et al. Radiometric characterization of field radiometers in support of the 1997 Lunar Lake, Nevada, experiment to determine surface reflectance and top-of-atmosphere radiance. *Remote Sens. Environ.* 2001, 77, 367–376. [CrossRef]

- 80. Thome, K.; Schiller, S.; Conel, J.; Arai, K.; Tsuchida, S. Results of the 1996 Earth Observing System vicarious calibration joint campaign at Lunar Lake Playa, Nevada (USA). *Metrologia* **1998**, *35*, 631–638. [CrossRef]
- Biggar, S.F.; Dinguirard, M.; Gellman, D.I.; Henry, P.; Jackson, R.D.; Moran, M.S.; Slater, P.N. Radiometric calibration of SPOT 2 HRV—A comparison of three methods. In Proceedings of the Calibration of Passive Remote Observing Optical and Microwave Instrumentation, Orlando, FL, USA, 1 August 1991; pp. 155–162.
- Sinyuk, A.; Holben, B.N.; Eck, T.F.; Giles, D.M.; Slutsker, I.; Korkin, S.; Schafer, J.S.; Smirnov, A.; Sorokin, M.; Lyapustin, A. The AERONET Version 3 aerosol retrieval algorithm, associated uncertainties and comparisons to Version 2. *Atmos. Meas. Tech.* 2020, 13, 3375–3411. [CrossRef]
- Dubovik, O.; Smirnov, A.; Holben, B.N.; King, M.D.; Kaufman, Y.J.; Eck, T.F.; Slutsker, I. Accuracy assessments of aerosol properties retrieved from Aerosol Robotic Network (AERONET) Sun and sky radiance measurments. *J. Geophys. Res.* 2000, 105, 9791–9806. [CrossRef]
- 84. Dubovik, O.; King, M.D. A flexible inversion algorithm for retrieval of aerosol optical properties from Sun and sky radiance measurements. *J. Geophys. Res.* 2000, *105*, 20673–20696. [CrossRef]
- 85. Chance, K.; Kurucz, R.L. An improved high-resolution solar reference spectrum for earth's atmosphere measurements in the ultraviolet, visible, and near infrared. *J. Quant. Spectrosc. Radiat. Transf.* **2010**, *111*, 1289–1295. [CrossRef]
- Thome, K.J.; Arai, K.; Tsuchida, S.; Biggar, S.F. Vicarious Calibration of ASTER via the Reflectance-Based Approach. *IEEE Trans. Geosci. Remote Sens.* 2008, 46, 3285–3295. [CrossRef]
- Leisso, N.P.; Thome, K.J.; Czapla-Myers, J.S. Validation of the onboard radiometric calibration of the GOES I-M visible channel by reflectance-based vicarious methods. In Proceedings of the Atmospheric and Environmental Remote Sensing Data Processing and Utilization III: Readiness for GEOSS, San Diego, CA, USA, 20 September 2007; pp. 668404–668410.
- 88. Thome, K.J.; Helder, D.L.; Aaron, D.; Dewald, J.D. Landsat-5 TM and Landsat-7 ETM+ absolute radiometric calibration using the reflectance-based method. *Geosci. Remote Sens. IEEE Trans.* **2004**, *42*, 2777–2785. [CrossRef]
- Anderson, N.J.; Biggar, S.F.; Burkhart, C.; Thome, K.J.; Mavko, M.E. Bi-directional Calibration Results for the Cleaning of Spectralon Reference Panels. In Proceedings of the Earth Observing Systems VII, Seattle, WA, USA, 24 September 2002; pp. 201–210.
- Biggar, S.F.; Labed, J.F.; Santer, R.P.; Slater, P.N.; Jackson, R.D.; Moran, M.S. Laboratory calibration of field reflectance panels. In Proceedings of the Recent Advances in Sensors, Radiometry, and Data Processing for Remote Sensing, Orlando, FL, USA, 12 October 1988; pp. 232–240.
- Jackson, R.D.; Susan Moran, M.; Slater, P.N.; Biggar, S.F. Field calibration of reference reflectance panels. *Remote Sens. Environ.* 1987, 22, 145–158. [CrossRef]
- 92. Kempen, T.; Rotmans, T.; Hees, R.; Bruegge, C.; Fu, D.; Hoogeveen, R.W.M.; Pongetti, T.; Rosenberg, R.; Aben, I. Vicarious Calibration of the TROPOMI-SWIR module over the Railroad Valley playa. *EGUsphere* **2023**, 1–32. [CrossRef]
- 93. Bruegge, C.J.; Coburn, C.; Elmes, A.; Helmlinger, M.C.; Kataoka, F.; Kuester, M.; Kuze, A.; Ochoa, T.; Schaaf, C.; Shiomi, K.; et al. Bi-Directional Reflectance Factor Determination of the Railroad Valley Playa. *Remote Sens.* **2019**, *11*, 2601. [CrossRef]
- 94. Slater, P.N.; Biggar, S.F.; Palmer, J.M.; Thome, K.J. Unified approach to absolute radiometric calibration in the solar-reflective range. *Remote Sens. Environ.* 2001, 77, 293–303. [CrossRef]
- 95. Thome, K.; Smith, N.; Scott, K. Vicarious calibration of MODIS using Railroad Valley Playa. In IGARSS 2001. Scanning the Present and Resolving the Future. In Proceedings of the IEEE 2001 International Geoscience and Remote Sensing Symposium, Sydney, NSW, Australia, 9-13 July 2001; pp. 1209–1211.
- Thome, K.J.; Crowther, B.G.; Biggar, S.F. Reflectance- and Irradiance-based Calibration of Landsat 5 Thematic Mapper. Can. J. Remote Sens. 1997, 23, 309–317. [CrossRef]
- Thome, K.J.; Gellman, D.I.; Parada, R.J.; Biggar, S.F.; Slater, P.N.; Moran, M.S. In-flight radiometric calibration of Landsat-5 Thematic Mapper from 1984 to present. In Proceedings of the Recent Advances in Sensors, Radiometric Calibration, and Processing of Remotely Sensed Data, Orlando, FL, USA, 15 November 1993; pp. 126–130.
- Thome, K.; D'Amico, J.; Hugon, C. Intercomparison of Terra ASTER, MISR, and MODIS, and Landsat-7 ETM+. In Proceedings of the IEEE International Conference on Geoscience and Remote Sensing Symposium, IGARSS 2006, Denver, CO, USA, 31 July–4 August 2006; pp. 1772–1775.
- Biggar, S.F.; Slater, P.N.; Gellman, D.I. Uncertainties in the in-flight calibration of sensors with reference to measured ground sites in the 0.4–1.1 μm range. *Remote Sens. Environ.* 1994, 48, 245–252. [CrossRef]
- Biggar, S.F.; Gellman, D.I.; Slater, P.N. Improved Evaluation of Optical Depth Components from Langley Plot Data. *Remote Sens. Environ.* 1990, 32, 91–101. [CrossRef]
- 101. Gellman, D.I.; Biggar, S.F.; Slater, P.N.; Bruegge, C.J. Calibrated intercepts for solar radiometers used in remote sensor calibration. In Proceedings of the Calibration of Passive Remote Observing Optical and Microwave Instrumentation, Orlando, FL, USA, 1 August 1991; pp. 175–180.
- 102. Thome, K.J.; Herman, B.M.; Reagan, J.A. Determination of Precipitable Water from Solar Transmission. *J. Appl. Meteorol.* **1992**, 31, 157–165. [CrossRef]
- Helder, D.; Thome, K.; Aaron, D.; Leigh, L.; Czapla-Myers, J.; Leisso, N.; Biggar, S.; Anderson, N. Recent surface reflectance measurement campaigns with emphasis on best practices, SI traceability and uncertainty estimation. *Metrologia* 2012, 49, S21. [CrossRef]

- 104. Thome, K.; Wenny, B.; Czapla-Myers, J.; Anderson, N.; Salehi, F. Solar radiation based calibration results from an ultra-portable field transfer radiometer used in vicarious calibrations. In Proceedings of the SPIE Optical Engineering + Applications, San Diego, CO, USA, 4 August 2021.
- 105. Thome, K.; Wenny, B.; Anderson, N.; McCorkel, J.; Czapla-Myers, J.; Biggar, S. Ultra-portable field transfer radiometer for vicarious calibration of earth imaging sensors. *Metrologia* **2018**, *55*, S104. [CrossRef]
- Thome, K.; Czapla-Myers, J.; Wenny, B.; Anderson, N. Calibration and use of an ultra-portable field transfer radiometer for automated vicarious calibration. In Proceedings of the Earth Observing Systems XXII, San Diego, CO, USA, 5 September 2017; pp. 181–191.
- Anderson, N.; Thome, K.; Czapla-Myers, J.; Biggar, S. Design of an ultra-portable field transfer radiometer supporting automated vicarious calibration. In Proceedings of the Earth Observing Systems XX, San Diego, CA, USA, 10–13 August 2015; Volume 9607, pp. 40–48.
- 108. Micijevic, E.; Haque, M.O.; Mishra, N. Radiometric calibration updates to the Landsat collection. In Proceedings of the Earth Observing Systems XXI, San Diego, CO, USA, 19 September 2016.
- Wulder, M.A.; Loveland, T.R.; Roy, D.P.; Crawford, C.J.; Masek, J.G.; Woodcock, C.E.; Allen, R.G.; Anderson, M.C.; Belward, A.S.; Cohen, W.B.; et al. Current status of Landsat program, science, and applications. *Remote Sens. Environ.* 2019, 225, 127–147. [CrossRef]
- 110. Czapla-Myers, J.; Woolliams, E. Uncertainty Analysis Statement—RVUS. 2022. Available online: www.radcalnet.org (accessed on 1 January 2024).
- 111. Wenny, B.N.; Thome, K. Look-up table approach for uncertainty determination for operational vicarious calibration of Earth imaging sensors. *Appl. Opt.* **2022**, *61*, 1357–1368. [CrossRef] [PubMed]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.